Author's response to reviews

Title: Most bowel symptoms do not indicate colorectal cancer and polyps: a systematic review

Authors:

Barbara-Ann Adelstein (b.adelstein@unsw.edu.au)
Petra Macaskill (petra.macaskill@sydney.edu.au)
Siew F Chan (siewc@health.usyd.edu.au)
Peter H Katelaris (peter.katelaris@sydney.edu.au)
Les Irwig (les.irwig@sydney.edu.au)

Version: 4 Date: 26 May 2011

Author's response to reviews:

Dear Editor

Re: MS ID: 1140363874550825

Most bowel symptoms do not indicate colorectal cancer and polyps: a systematic review

Authors: Barbara-Ann Adelstein, Petra Macaskill, Siew F Chan, Peter H Katelaris and Les Irwig

Thank you for your email requesting further changes to the manuscript. In response to this:

• All additional files are now referred to as such, in the manuscript text and in the files themselves.

• I have removed the repeated manuscript title from the abstract and from above ‘Background’.

• I have cropped the figures closely (and submitted them as pdfs, not eps files).

I hope that this resolves all proofing issues; please let me know if you require any further changes.

Yours sincerely

Barbara-Ann Adelstein